Author’s response to reviews

Title: A carcinoid heart disease revealed by cyanosis with both right and left valvular involvement: a case report.

Authors:

Khadija Khay (khadija.khayi5@gmail.com)
Salim Arous (arous.salim@hotmail.fr)
Tarik Bentaoune (bentarik@gmail.com)
Abdenasser Drighil (sdrighil@hotmail.fr)
Rachida Habbal (hrachida@hotmail.fr)

Version: 1 Date: 06 Jan 2018

Author’s response to reviews:

Dear reviewers,

- The background is rewritten, and the discussion is modified with more references.
- The document is written in past except of the discussion.
- Figure 1 is changed.

The treatment is more detailed

Cordially